%% Math 131, Fall, 2003
\documentclass[12pt]{article}
\usepackage{latexsym}

\pagestyle{empty}

\setlength{\textwidth}{6.5in}
\setlength{\oddsidemargin}{0in}
%% \setlength{\topmargin}{-.5in}
\setlength{\textheight}{8.25in}

\def\ds{\displaystyle}
\def\R{\textbf{R}}
\def\Item{\item{$\Box$}\quad}

\def\be{\begin{enumerate}}
\def\ee{\end{enumerate}}

\def\bd{\begin{description}}
\def\ed{\end{description}}

\def\Answer#1{$\phantom{.}$\hfill
        #1\quad\fbox{$\phantom{\Big|Answer}$}} 

\begin{document}

\noindent{\bf Name}:\ \vbox{\hrule width 3in}

\begin{center}
\textbf{Holy Cross College, Fall Semester, 2003} \\
\textbf{Math 131, Quiz \#1 Make-Up} \\
Professor Hwang \\
\end{center}

\be
\item Let $f(t)=P_0\,a^t$, and assume that $f(-1)=2$~and $f(3)=18$.

  \bd
  \item{(a)} Find $P_0$~and $a$.

    \vfil

    \Answer{$P_0=$}\Answer{$a=$}

    \begin{center}
      \hrule
    \end{center}

  \item{(b)} Find the average rate of change of~$f$ on the
    interval~$[1,3]$.

    \vfil
    \Answer{Average rate of change on~$[1,3]$ is}
  \ed
\ee
\eject

\end{document}
